Author's response to reviews

Title: Rationale, design, and protocol for the prevention of low back pain in the military (POLM) trial (NCT00373009)

Authors:

Steven Z George (sgeorge@phhp.ufl.edu)
John D Childs (childsjd@sbcglobal.net)
Deydre S Teyhen (dteyhen@sbcglobal.net)
Samuel S Wu (samwu@biostat.ufl.edu)
Alison C Wright (alison.wright@us.army.mil)
Jessica L Dugan (jessie@evidenceinmotion.com)
Michael E Robinson (merobin@ufl.edu)

Version: 3 Date: 9 September 2007

Author's response to reviews: see over
Thank you for accepting our manuscript, pending formatting changes. Our responses are listed below each requested change.

Requested Formatting Changes:

Please ensure that the title (in sentence case) and the list of authors' names are identical in the submission system and the manuscript (Jessie/Jessica).

**This was done. Jessica is the preferred name of the author.**

EndNote or Ref Manager references - Please ensure that the data in your Ref Manager software is entered correctly. This is to ensure that the references are displayed correctly.

**References were checked for accuracy. One in press reference was updated.**

Please remove the website from the acknowledgements section (page25).

**This was done.**

Please ensure that you format your reference list before re-submitting your revised manuscript.

Currently, the reference list does not have the same sequence as the extracted reference list (reference 7 onwards).

**There were duplicate references at #5 and #8, and #4 and #7. These were consolidated. All other references were checked for accuracy, and appeared to be displayed correct.**

Please remove the brackets from the numbers in the references section, and align the header to the left.

**Both of these were done.**

Please remove the following text from the table legends for tables 1 and 2 "Table Legend".

**This was done, the same text was also removed from Table 3.**
Thank you for accepting our manuscript for publication. Please contact us if there are any more issues that need to be resolved before going to press.

Sincerely,

John D. Childs, PT, PhD, MBA
Assistant Professor and Director of Research
US Army-Baylor University Doctoral Program in Physical Therapy